Author's response to reviews

Title: Comparison of AIMS2-SF, WOMAC, x-ray and a global physician assessment in order to approach quality of life in patients suffering from osteoarthritis

Authors:

- Thomas TR Rosemann (thomas_rosemann@med.uni-heidelberg.de)
- Stefanie SJ Joos (stefanie_joos@med.uni-heidelberg.de)
- Thorsten TK Koerner (thorsten_koerner@med.uni-heidelberg.de)
- Joachim JS Szecsenyi (joachim_szecsenyi@med.uni-heidelberg.de)
- Gunter GL Laux (gunter_laux@med.uni-heidelberg.de)

Version: 4 Date: 7 December 2005

Dear Editorial Production Team,

according to the requests mentioned in your email we performed the following changes:

Title & Abstract - We ensure that these are identical in the manuscript and in the submission system.

Author list - The list of authors in the manuscript is now written exactly as they are in the submission system.

Figure title - We skipped the title of the image files and cropped them as closely as possible.

Sincerely

Thomas Rosemann